University of Nebraska - Lincoln DigitalCommons@University of Nebraska - Lincoln

Civil and Environmental Engineering Faculty Publications

Civil and Environmental Engineering

2022

Geospatial Distribution of Age-Adjusted Incidence of the Three Major Types of Pediatric Cancers and Waterborne Agrichemicals in Nebraska

Balkissa S. Ouattara

Jagadeesh Puvvula

Azar Abadi

Siddhi Munde

Alan S. Kolok

See next page for additional authors

Follow this and additional works at: https://digitalcommons.unl.edu/civilengfacpub

Part of the Civil and Environmental Engineering Commons

This Article is brought to you for free and open access by the Civil and Environmental Engineering at DigitalCommons@University of Nebraska - Lincoln. It has been accepted for inclusion in Civil and Environmental Engineering Faculty Publications by an authorized administrator of DigitalCommons@University of Nebraska - Lincoln.

Authors

Balkissa S. Ouattara, Jagadeesh Puvvula, Azar Abadi, Siddhi Munde, Alan S. Kolok, Shannon Bartelt-Hunt, Jesse E. Bell, Christopher S. Wichman, and Eleanor Rogan





GeoHealth

RESEARCH ARTICLE

10.1029/2021GH000419

Key Points:

- The incidence of pediatric brain and other central nervous system cancers was higher than the national average in many counties in Nebraska
- Atrazine concentration in groundwater was associated with pediatric cancers
- Nitrate concentration in groundwater was associated with pediatric cancers

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to: B. S. Ouattara,

balkissa.ouattara@unmc.edu

Citation:

Ouattara, B. S., Puvvula, J., Abadi, A., Munde, S., Kolok, A. S., Bartelt-Hunt, S., et al. (2022). Geospatial distribution of age-adjusted incidence of the three major types of pediatric cancers and waterborne agrichemicals in Nebraska. *GeoHealth*, 6, e2021GH000419. https:// doi.org/10.1029/2021GH000419

Received 4 MAR 2021 Accepted 1 FEB 2022

Author Contributions:

Conceptualization: Balkissa S. Ouattara, Jagadeesh Puvvula, Azar Abadi, Alan S. Kolok, Shannon Bartelt-Hunt, Eleanor Rogan Data curation: Balkissa S. Ouattara, Siddhi Munde Formal analysis: Balkissa S. Ouattara, Christopher S. Wichman

Methodology: Balkissa S. Ouattara, Azar Abadi, Alan S. Kolok, Shannon Bartelt-Hunt, Eleanor Rogan Software: Balkissa S. Ouattara Supervision: Eleanor Rogan

© 2022 The Authors. GeoHealth published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Geospatial Distribution of Age-Adjusted Incidence of the Three Major Types of Pediatric Cancers and Waterborne Agrichemicals in Nebraska

Balkissa S. Ouattara¹, Jagadeesh Puvvula¹, Azar Abadi¹, Siddhi Munde¹, Alan S. Kolok², Shannon Bartelt-Hunt³, Jesse E. Bell¹, Christopher S. Wichman⁴, and Eleanor Rogan¹

¹Department of Environmental, Agricultural and Occupational Health, College of Public Health, University of Nebraska Medical Center, Omaha, NE, USA, ²Idaho Water Resources Research Institute, University of Idaho, Moscow, ID, USA, ³Department of Civil and Environmental Engineering, College of Engineering, University of Nebraska-Lincoln, Omaha, NE, USA, ⁴Department of Biostatistics, College of Public Health, University of Nebraska Medical Center, Omaha, NE, USA

Abstract This study was conducted to examine, at the county level, the relationship between pediatric cancer incidence rate and atrazine and nitrate mean concentrations in surface and groundwater. A negative binomial regression analysis was performed to investigate the association between central nervous system (CNS) tumors, leukemia, lymphoma, and atrazine and nitrate mean concentrations in groundwater. The age-adjusted brain and other CNS cancer incidence was higher than the national average in 63% of the Nebraska counties. After controlling for the counties socio-economic status and nitrate concentrations in groundwater, counties with groundwater atrazine concentrations above 0.0002 µg/L had a higher incidence rate for pediatric cancers (brain and other CNS, leukemia, and lymphoma) compared to counties with groundwater atrazine concentrations in the reference group (0.0000–0.0002 µg/L). Additionally, compared to counties with groundwater nitrate concentrations between 0 and 2 mg/L (reference group), counties with groundwater nitrate concentrations between 2.1 and 5 mg/L (group 2) had a higher incidence rate for pediatric brain and other CNS cancers (IRR = 8.39; 95% CI: 8.24–8.54), leukemia (IRR = 7.35; 95% CI: 7.22–7.48), and lymphoma (IRR = 5.59; CI: 5.48-5.69) after adjusting for atrazine groundwater concentration and the county socioeconomic status. While these findings do not indicate a causal relationship, because other contaminants or cancer risk factors have not been accounted for, they suggest that atrazine and nitrate may pose a risk relative to the genesis of pediatric brain and CNS cancers, leukemia, and lymphoma.

Plain Language Summary The rate of pediatric cancers in Nebraska is currently among the five highest in the United States. Ninety-two percent (92%) of Nebraska state's total land area is used for agriculture (farming and ranching). It is challenging to establish childhood cancer causes because only one in 20 cases are related to heredity. Statistical tools were used to investigate the relationship between the exposure to nitrate and atrazine in surface and groundwater and childhood cancers in Nebraska. Nebraska counties where atrazine or nitrate levels were elevated, reported more childhood cancers than counties with lower levels of these chemicals. These results suggest that different agricultural activities across the state might present a risk to develop certain pediatric cancers.

1. Introduction

Nebraska is primarily an agricultural state, with 92% of the total land area used for farming and ranching (Nebraska Department of Agriculture and USDA NASS, Nebraska Field Office, 2020). The widespread use of agrichemicals such as atrazine and nitrate is common across many Midwestern states, including Nebraska. Atrazine, a triazine herbicide, is the second most used pesticide in Nebraska for leafy weed control (Wieben, 2019), and in 2017, more than 3,357 tons of atrazine were applied to Nebraska cornfields (Wieben, 2019). Nitrate is used as a fertilizer to enhance crop growth. In 2018, about 742,000 tons of nitrogen-containing fertilizers were applied to Nebraska cornfields (Nebraska Department of Agriculture, N.D.; USDA ERS, 2019).

These agrichemicals can contaminate surface water (SW) and groundwater (GW), the latter being an essential drinking water source for more than 85% of all Nebraskans (Nebraska Department of Environmental Quality, 2020). Indeed, in Nebraska, surface and groundwater concentrations of nitrate and atrazine have been found in some locations to exceed the United States (US) Environmental Protection Agency (EPA) maximum contaminant



Validation: Balkissa S. Ouattara, Eleanor Rogan Writing – original draft: Balkissa S. Ouattara, Azar Abadi Writing – review & editing: Balkissa S. Ouattara, Jagadeesh Puvvula, Siddhi Munde, Alan S. Kolok, Shannon Bartelt-Hunt, Jesse E. Bell, Eleanor Rogan limits (MCL) of 10 mg/L for nitrate as nitrogen and three (3) µg/L for atrazine (EPA, 2020a; Nebraska Department of Environmental Quality, 2020). Under the Safe Drinking Water Act (SDWA), the EPA has set up, for more than 90 contaminants, health-based drinking water standards and regulations. However, the SDWA does not cover private wells that serve fewer than 25 people (EPA, 2020b). It then falls under the responsibility of domestic well owners to ensure the quality and safety of their drinking water (DeSimone et al., 2009; EPA, 2020b).

A study that assessed the water quality of 2100 domestic wells in 48 states located in major aquifers of the United States reported that approximately one in five wells contained at least one contaminant with a concentration above the EPA maximum contaminant limit. The contaminants found at elevated concentrations included metals, radionuclides, and nitrate (DeSimone et al., 2009).

While research involving atrazine has been controversial, some studies have identified an association between atrazine exposure and adverse health outcomes in humans. For example, maternal atrazine exposure was associated with birth defects such as male genital malformations and gastroschisis (Agopian et al., 2013; Waller et al., 2010). Moreover, atrazine exposure has been associated with cancers such as pediatric leukemia and reproductive cancers (Booth et al., 2015; Carozza et al., 2008; Fan et al., 2007; Freeman et al., 2011; Malagoli et al., 2016). Other studies found no evidence of association between atrazine exposure and cancers. For instance, research among licensed pesticide applicators found no increase in overall cancer risk among atrazine users when comparing the highest and lowest exposure categories (Rusiecki et al., 2004). Because of these inconclusive results, the International Agency for Research on Cancer (IARC) lists atrazine in Group 3: not classifiable about its carcinogenicity to humans due to conflicting experimental results. Likewise, the EPA has concluded that human and animal evidence was not sufficient to consider atrazine carcinogenic (Boffetta et al., 2013).

In contrast to atrazine, the association between relatively high nitrate concentration in water and adverse health impacts is well established. For example, in 1945, infant methemoglobinemia was associated with elevated nitrate concentrations in drinking water (Du et al., 2007; Monti et al., 2019). Since then, relatively high nitrate concentrations in drinking water have been associated with many adverse health outcomes, including hypothyroidism (Aschebrook-Kilfoy et al., 2012), congenital anomalies (Brender & Weyer, 2016; Holtby et al., 2014), and malignant tumors such as colorectal, bladder, and kidney cancer (Fathmawati et al., 2017; Jones et al., 2016, 2017). Relative to carcinogenicity, ingested nitrate is reduced to nitrite, which interacts with amides or amines to form N-nitroso compounds that are established carcinogens and teratogens (Mensinga et al., 2003; Ward et al., 2018). Although the EPA has not yet classified nitrates as carcinogenic (EPA, 2007), IARC (2010), in its 94th volume monograph, stated that "ingested nitrate or nitrite under conditions that result in endogenous nitrosation is probably carcinogenic to humans (Group 2A)."

The incidence of pediatric cancers has been high in Nebraska (Corley et al., 2018) and above the national average (National Cancer Institute, 2017). Since only about 5% of all childhood cancers are hereditary, this suggests that other external factors such as environmental exposure to carcinogenic chemicals may play a prominent role in their etiology (NIH, 2020; Robin & Farmer, 2017). This study was conducted to examine, at the county level, the relationship between pediatric cancer incidence rate and atrazine and nitrate mean concentrations in surface and groundwater; the authors hypothesized that relatively higher concentrations of nitrate and atrazine in surface and groundwater in Nebraska are positively associated with higher pediatric cancer incidence rate.

2. Materials and Methods

2.1. Case Definition, Study Population, and Data Sources

2.1.1. Pediatric Cancer Data

Cases were defined as all children aged 0–19 years of age and diagnosed with brain and other CNS cancers, leukemia, and lymphoma recorded in the Nebraska Cancer Registry between 01 January 1987 and 31 December 2016. Based on the case definition, the at-risk population encompasses all the children (0–19 years of age) who lived in Nebraska from 1987 to 2016. Pediatric cancer data were obtained from the Nebraska Department of Health and Human Services (DHHS) Cancer Registry.

2.1.2. Water Quality Data

Atrazine and nitrate data came from monitoring well sampling and were retrieved from the water quality portal (National Water Quality Monitoring Council, 2021) and Nebraska Quality-Assessed Agrichemical Contaminant Database (University of Nebraska-Lincoln, 2020). We retrieved water quality data corresponding to our study period, 1987–2016. Monitoring wells' locations and sampling frequency are decided based on findings of high contaminant levels during random field testing. Many entities including the state's 23 Natural Resource Districts, performed groundwater monitoring in Nebraska to address contaminants like nitrate and atrazine in their jurisdiction (Nebraska Department of Environmental Quality, 2020).

Monitoring wells share similar geographical and geological (same aquifer for example) characteristics with drinking water wells, and they are constructed for the purposes of water quantity and quality data collection and serve as a proxy for drinking water quality in a given geographic area (Nebraska Department of Environmental Quality, 2020).

2.1.3. GIS and Other Data

The Nebraska state and county boundary shapefiles were extracted from the United States Census Bureau (2019). The 2010 U.S decennial census data and Nebraska county populations were obtained from the National Historical Geographic Information System database (Manson et al., 2021). The Centers for Disease Control and Prevention (CDC)'s social vulnerability index (SVI) was available for download from the Centers for Diseases Control and Prevention and Prevention (ATSDR, 2016).

2.2. Data Analysis

2.2.1. Age-Adjusted Incidence

The age-adjusted incidence for each Nebraska county and the state as a whole were determined by first calculating the crude incidence according to the equation (National Cancer Institute, 2021):

Crude incidence = $[New cases_{county} / (Population at risk_{county} * Time of analysis)] * 100000$

The crude incidence was then used to determine the age-adjusted incidence according to the equation (National Cancer Institute, 2021):

Age-adjusted incidence = \sum crude incidence_{county} * Age distribution of standard population_{Age group}.

The ages were categorized into four groups: 0-4, 5-9, 10-14, and 15-19 years.

The age distribution of the standard population was obtained by dividing the population in any specific age group by the total U.S. 2010 standard population.

We used the U.S 2010 census population to calculate the age-adjusted incidence and as the offset variable in the negative binomial regression analysis for three reasons. First, we referred to published literature (Corley et al., 2018; Thorpe & Shirmohammadi, 2005) that conducted an analysis over an extended study period using a single census year. Second, childhood cancers although a leading cause of death in children are rare, and our approach to look at incidence at the county level (there are 93 counties in Nebraska) and to investigate subtypes of pediatric cancers make the event rarer and annual case counts very small. Finally, we divided our study period to a 10-year time series (1987–1996, 1997–2006, and 2007–2016) to match the three decennial censuses (1990, 2000, and 2010) that occurred during our study period. We found the same trend (decrease in crude incidence for all three cancer types) in both counties with rapid population growth and counties with less or no growth.

Counties with a total pediatric (0–19 years old) population of 200 or fewer people were excluded from the analysis. Out of the 93 Nebraska counties, we included 83 counties in the study. Ten counties (Arthur, Banner, Blaine, Grant, Hooker, Keya Pawa, Logan, Loup, McPherson, and Thomas) were excluded. The cutoff of 200 was based on our exploratory data analysis findings showing extreme incidence rates for the counties with a population of 200 or lower.

Classification of Mean Atrazine and Nitrate Concentrations in Categories

	Atrazine (µg/L)		Nitrate (mg/L)	
Categories	SW ^a	GW ^b	SW	GW
Group 1 (reference)	0.0000-0.1313	0.0000-0.0002	0.0000-2.0000	0.0000-2.0000
Group 2	0.1314-0.9473	0.0003-0.0213	2.1000-5.0000	2.1000-5.0000
Group 3	0.9474–2.8187	0.0214-0.0995	5.0100-10.0000	5.0100-10.0000
Group 4	2.8188-18.5750	0.0996-2.5118	10.0100-12.4200	10.0100-15.1500
internet a larger				

^aSW = surface water. ^bGW = groundwater.

Finally, using ArcGIS Pro version 2.4 (ESRI, 2019), the spatial distribution of the age-adjusted incidence of pediatric brain and other CNS cancers, leukemia, and lymphoma in Nebraska was compared to the national average.

2.3. Negative Binomial Regression Analysis

2.3.1. Dependent (Outcome) Variables

Both univariable and multivariable negative binomial regression analyses were conducted in SPSS (Statistical Package for the Social Sciences; IBM Corp, 2019), to identify predictors of the three most common pediatric cancer counts (brain and other CNS cancers, leukemia, and lymphoma) with offset for the county level pediatric population size. We built separate models with our three outcome variables, count of brain and other CNS cancers, count of leukemia, and count of lymphoma. Our full model was set as followed:

 $\ln(\text{counts}) = b0 + b1 * \text{atrazine } GW + b2 * \text{nitrate } GW + b3 * RPL_theme1 + offset(\ln(\text{population}))$

2.3.2. Independent Variables

The independent variables were the growing season mean nitrate and atrazine concentrations in groundwater during our study period. We chose to use the mean concentration instead of annual measurements according to the limitations of our data. First, we do not know the duration of exposure or the latency period between exposure and cancer diagnosis. Second, annual atrazine and nitrate measurements were not available for many counties. About 30% (25) of the counties included in the study have a yearly sampling frequency of less than one (Tables S1–S4 in Supporting Information S1) for surface and groundwater. Finally, our preliminary findings showed that nitrate or atrazine concentrations have not dramatically changed during the study period (the mean estimate changed by less than 5%). We assumed that all the children diagnosed with cancer during the study period had similar exposure regarding the contaminant levels. The growing season (months of May to October) was emphasized, as atrazine and nitrate are applied during the growing season, and their concentration in the water is expected to be higher with higher chances of exposure. We also used a covariate in the analysis: the CDC SVI for each county to account for the county socioeconomic status (RPL_Theme1). The CDC SVI assesses the relative social vulnerability of each county using different factors that may impact the health of the

 Table 2

 Most Common Pediatric Cancers Types in Nebraska (1987–2016) and the United States

Cancer type	Frequency in Nebraska (%)	Frequency in the United States (%)
Brain and other CNS	26	18
Leukemia	24	26
Lymphoma	16	14
Other types	34	42

population (ATSDR, 2016). Because socioeconomic status impact the health of the population (ATSDR, 2016). Because socioeconomic status impacts cancer incidence in general (Clegg et al., 2009; Garcia-Gil et al., 2014), we opted to control for the county socioeconomic status in our analysis. The atrazine and nitrate concentrations were classified into four groups (Table 1) using the quantile classification in ArcGIS, to determine automatic groups for atrazine concentrations in surface and groundwater. For nitrate concentrations in surface and groundwater. For nitrate concentrations in surface and groundwater, the function "manual intervals" in ArcGIS was used to set specific concentration ranges (ESRI, 2020). We selected 2 mg as the first cutoff for the nitrate groups because studies documented that nitrate concentrations above 2 mg/L are likely related to anthropogenic activities





Figure 1. (a) Mean groundwater atrazine concentration per county in Nebraska from 1987 to 2016. (b) Mean surface water atrazine concentration per county in Nebraska from 1987 to 2016.

such as the use of fertilizers (Nolan & Stoner, 2000; Rhoades et al., 2013). The 10 mg/L cutoff referred to the EPA maximum contaminant limit.

3. Results

3.1. Descriptive Statistics

3.1.1. Pediatric Cancers

Among the 2559 pediatric cancer cases reported in the Nebraska cancer registry from 1987 to 2016, 13 types of cancers were identified. Brain and other CNS cancers were the most represented with 26% (665/2559) of all cases; followed by leukemia, 24.4% (625/2559); and lymphoma, 16% (405/2559). Table 2 compares the most predominant pediatric cancer types in Nebraska and the United States (ACCO, 2014).

3.1.2. Nitrate and Atrazine Concentration

Growing seasons groundwater (GW) and SW mean atrazine and nitrate concentrations are represented for counties included in the studies and where measurements were completed (Figures 1 and 2).

In addition to counties not included in our study, some counties were not sampled at all for SW atrazine and nitrate during our study period. All quartiles of mean groundwater atrazine and nitrate concentration are significantly represented across the counties included in the study. In contrast, for SW atrazine and nitrate, the first quartile (lowest concentration) is the most represented among the counties.

Descriptive statistics for the 10 counties that were most frequently and least frequently sampled during our study period are shown in Tables 3–6. For full list of counties please refer Tables S1-S6 in Supporting Information S1.



Figure 2. (a) Mean groundwater nitrate concentration per county in Nebraska from 1987 to 2016. (b) Mean groundwater nitrate concentration per county in Nebraska county, 1987–2016.



Descriptive Statistics of Groundwater Atrazine for Selected Counties With Lower and Higher Annual Sampling Frequency

County	Sampling frequency ^a	Total wells ^b	Minimum concentration ^c	Mean concentration ^c	Maximum concentration ^c
Nemaha	<1	2	0.00	0.00	0.00
Hooker	<1	3	0.00	0.00	0.00
Otoe	<1	7	0.00	0.05	0.10
Johnson	<1	11	0.00	0.02	0.09
Webster	<1	12	0.00	0.07	0.20
Saunders	13	125	0.00	0.10	1.64
Cass	14	69	0.00	0.05	0.50
Lancaster	26	184	0.00	0.02	0.45
York	27	66	0.00	0.73	2.08
Buffalo	223	109	0.00	1.02	3.03

^aAnnual average sampling frequency. ^bNumber of monitoring wells sampled from 1987 to 2016. ^cConcentration in µg/L.

In general, the number of wells available in a county was related to the sampling frequency.

For groundwater atrazine, the average concentration in all counties was below the EPA MCL. In contrast, 16% of all groundwater nitrate measurements were above the MCL of 10 mg/L.

3.2. Longitudinal Analysis

3.2.1. Pediatric Cancer Data

We divided the health data set into three 10-year periods. Compared to the first 10 years (1987–1996) and third 10-year period (2007–2016), CNS cancer and lymphoma crude incidences were higher in most counties during the second 10-year period (1997–2006; Figures 3a-3c).

3.2.2. Water Quality Data

Table 4

We split the water quality data into three 10-year periods. Compared to the first 10 years (1987–1996) and the third 10-year period (2007–2016), atrazine in groundwater during the years 1997–2006 were more consistently measured and had higher concentrations (Figure 4a).

.

Descriptive statistics of surface water Atrazine for Selected Counties with Lower and Higher Annual Sampling Frequency					
County	Sampling frequency ^a	Total wells ^b	Minimum concentration ^c	Mean concentration ^c	Maximum concentration ^c
Clay	<1	1	3.30	3.85	4.40
Custer	<1	1	0.00	0.06	0.11
Hamilton	<1	2	5.15	18.58	32.00
Franklin	<1	1	0.54	1.14	1.36
Phelps	<1	2	0.16	0.23	0.31
Richardson	21	18	0.00	1.69	37.60
Dodge	25	27	0.01	1.80	30.80
Sarpy	29	18	0.03	1.80	26.00
Lancaster	36	44	0.00	4.69	224.00
Douglas	54	51	0.01	1.07	18.42

^aAnnual average sampling frequency. ^bNumber of monitoring wells sampled from 1987 to 2016. ^cConcentration in µg/L.



Descriptive Statistics of Groundwater Nitrate for Selected Counties With Lower Higher and Annual Sampling Frequency

County	Sampling frequency ^a	Total wells ^b	Minimum concentration ^c	Mean concentration ^c	Maximum concentration ^c
Hooker	1	11	0.00	0.58	1.94
Valley	3	71	0.00	3.51	9.30
Hayes	4	72	1.65	3.14	5.34
Otoe	6	83	0.00	7.04	27.80
Frontier	6	119	3.00	5.00	8.24
Platte	175	912	0.00	12.66	36.20
Box Butte	183	1133	1.65	3.38	10.18
Antelope	195	386	2.00	9.45	16.52
Buffalo	232	411	0.00	11.91	40.00
Holt	280	789	1.07	12.73	31.75

^aAnnual average sampling frequency. ^bNumber of monitoring wells sampled from 1987 to 2016. ^cConcentration in mg/L.

Regarding groundwater nitrate, measurements were less frequent but higher during the same period (1997–2006; Figure 4b).

3.3. Age-Adjusted Incidence of Pediatric Cancers in Nebraska and the National Average

The age-adjusted incidence for pediatric brain and other CNS cancers in Nebraska was 4.42 per 100,000 population between 1987 and 2016 (Table 7). This incidence was higher than the national average age-adjusted incidence for pediatric brain and other CNS cancers reported to be 3.16 per 100,000 population in average between 1999 and 2016 (U.S. Cancer Statistics Working Group, 2020). The incidence of leukemia (3.67 per 100,000 persons) and lymphoma (2.72 per 100,000 persons) was lower in Nebraska than the national average.

3.4. Geospatial Analysis

Table 6

The incidence of the three major types of pediatric cancers is shown on a map of Nebraska with counties delineated (Figures 5–7). For each cancer type, we represented side by side two maps, one with four colors on the map representing pediatric cancer incidence in quartiles with the first quartile incidence below the national average. The second map in each panel has only two colors (yellow and green). The green color represents counties with

Descriptive Statistics of Surface Water Nitrate for Selected Counties With Lower and Annual Sampling Frequency					
County	Sampling frequency ^a	Total wells ^b	Minimum concentration ^c	Mean concentration ^c	Maximum concentration ^c
Cedar	<1	1	0.12	0.12	0.12
Nemaha	<1	1	1.23	1.23	1.23
Dawes	<1	2	0.04	0.22	0.41
Madison	<1	1	0.00	9.50	18.99
Otoe	<1	2	0.09	0.11	0.13
Stanton	<1	1	0.00	0.00	0.00
Lancaster	19	27	0.07	2.34	8.52
Dodge	23	6	0.16	4.48	9.15
Sarpy	32	7	0.00	1.06	3.98
Douglas	52	31	0.00	1.71	5.79

^aAnnual average sampling frequency. ^bNumber of monitoring wells sampled from 1987 to 2016. ^cConcentration in mg/L.





Figure 3. (a) Pediatric central nervous system crude incidence in Nebraska counties over three decades. (B) Pediatric lymphoma crude incidence in Nebraska counties over three decades.

incidence above the national average and the yellow counties with incidence below the national average. Counties excluded from the analysis are left blank.

Relative to the national average, the age-adjusted incidence of pediatric brain and other CNS cancers (Figure 5a) is higher in 63% (54/86) of the Nebraska counties. In 41% (35/86) and 43% (38/86) of Nebraska counties; the incidence of pediatric cancers is higher than the national average, respectively, for leukemia (Figure 6a) and lymphoma (Figure 7a).

3.5. Analysis of the Relation Between Pediatric Cancers (Brain and Other CNS, Leukemia, and Lymphoma) and Agrichemicals in Groundwater (Nitrate and Atrazine)

As mentioned above, the contaminant concentrations in our data set are a proxy for their levels in drinking water. Thereby, we focused the regression analysis on the groundwater concentrations of atrazine and nitrate. Table 8 represents the results of the negative binomial multivariable analysis.



Figure 4. (a) Longitudinal analysis of groundwater atrazine in Nebraska counties. (b) Longitudinal analysis of groundwater nitrate in Nebraska counties.

3.5.1. Brain and Other CNS Cancers

Compared to counties with groundwater atrazine concentration in group 1 (reference group), counties with groundwater atrazine concentration in groups 2, 3, and 4 have brain and other CNS cancers incidence rate ratio of 6.46 (95% CI: 6.35–6.57), 11.20 (95% CI: 11.01–11.39), and 13.50 (95% CI: 13.26–13.74), respectively.

Regarding nitrate, keeping all other variables (atrazine concentration in groundwater and RPL_theme1) constant, counties with groundwater concentration in group 2 have a higher incidence rate of pediatric brain and other CNS cancers than counties with groundwater nitrate concentration in the reference group 1 (IRR = 8.39; 95% CI: 8.24–8.54).

3.5.2. Leukemia

After adjusting for all other variables (nitrate concentration in groundwater and RPL_theme1) in the model, counties with groundwater atrazine concentration in groups 2, 3, and 4 have leukemia incidence rates of 3.90 (95% CI: 3.83–3.96), 6.28 (95% CI: 6.18–6.38), and 6.22 (95% CI: 6.12–6.32) times higher than leukemia incidence rate for counties with groundwater atrazine concentration in the reference group (group1), respectively. Regarding nitrate, keeping all other variables (atrazine concentration in group 2 have a higher incidence rate of leukemia than counties with groundwater nitrate concentration in group 1 (IRR = 7.35; 95% CI: 7.22–7.48).

3.5.3. Lymphoma

The incidence rate of lymphoma in counties with groundwater atrazine concentration in groups 2, 3, and 4 were 7.81 (95% CI: 7.67–7.95), 12.76 (95% CI: 12.53–13.00), and 12.79 (95% CI: 12.55–13.04) times higher than the incidence rate for the reference group.

Regarding nitrate, keeping all other variables constant, counties with groundwater nitrate concentration in group 2 have a higher incidence rate of lymphoma than counties with groundwater nitrate concentration in group 1 (IRR = 5.59; CI: 5.48-5.69).



Age-Adjusted Incidence of Pediatric Cancers in Nebraska and the United States

Cancer type	Nebraska age-adjusted incidence (per 100,000)	National age-adjusted incidence (per 100,000)
Brain and other CNS cancers	4.42	3.16
Leukemia	3.67	4.66
Lymphoma	2.72	2.65

4. Discussion

The relationship between nitrate and atrazine mean concentrations in groundwater with the three most prevalent pediatric cancer types in Nebraska was investigated through an ecological study that accounts for two critical waterborne agrichemicals used in Nebraska. We found some associations between atrazine and nitrate concentrations with pediatric cancer incidence rates.

These findings add to the growing number of studies that have observed an association between atrazine levels in water and increased cancer incidence (Booth et al., 2015; Carozza et al., 2008; Fan et al., 2007; Freeman et al., 2011; Malagoli et al., 2016). For example, Freeman et al. (2011), with a small sample size, found an increased risk of ovarian cancer among female applicators of atrazine compared to female nonapplicators. Furthermore, higher incidence

rates of pediatric leukemia were observed in Illinois counties with greater than the median acreage of corn (Booth et al., 2015). Atrazine is one of the most common herbicides used in corn production; thus Booth et al. (2015) in their study implied an association between atrazine use and pediatric leukemia (RR Leukemia = 2.09, 95% CI = 1.31-3.32). Similarly, research showed an association between residence at the time of diagnosis in agriculturally intense areas and increased childhood cancer incidence. The assumption was that agriculturally intensive areas used many pesticides, including atrazine (Carozza et al., 2008). The risk of pediatric leukemia increased with arable crop production dominated by the use of atrazine, as suggested by Malagoli et al. (2016).

Furthermore, several preclinical studies in animal models reported that embryonic exposure to low-dose atrazine is associated with alterations in embryonic genetic expression that promote carcinogenesis (Gely-Pernot et al., 2017; Horzmann et al., 2018; Weber et al., 2013). For instance, zebrafish embryo exposure to different doses (0.3, 3, and 30 ppb) of atrazine resulted in genetic and molecular alterations, which were associated with teratogenicity and carcinogenesis in these models (Weber et al., 2013).

However, many other studies did not find an association between atrazine exposure and increased cancer risk (Rusiecki et al., 2004; Sathiakumar et al., 2011). Although Rhoades et al. (2013) in their research found that the odds of developing Non-Hodgkin lymphoma (NHL) were 2.9 times (CI: 1.1–7.4) higher in subjects exposed to both atrazine and nitrate in water, the study did not observe an association between NHL risk and nitrate or atrazine alone.

The present study also found that Nebraska counties with groundwater nitrate concentration in group 2 have higher incidence rates for all three major types of pediatric cancer than counties with groundwater nitrate concentration in group 1 (reference group). These findings suggest that intensive agriculture, the primary source of water contamination by nitrate, contributes to the excess rate of pediatric cancers in Nebraska. The current results will also enrich the body of evidence of a positive relationship between nitrate concentration and increased cancer risk in humans. Indeed, a significant number of studies have found an association between exposure to a relatively high nitrate concentration and the risk of developing cancer. Examples of such studies include case-con-



Figure 5. (a) Age-adjusted incidence (AAI) of pediatric brain and other CNS cancers per county in Nebraska from 1987 to 2016. (b) Age-adjusted incidence (AAI) of pediatric brain and other CNS cancers in Nebraska counties compared to the national average.





Figure 6. (a) Age-adjusted incidence (AAI) of pediatric leukemia per county in Nebraska from 1987 to 2016. (b) Age-adjusted incidence (AAI) of pediatric leukemia in Nebraska counties compared to the national average.

trol research conducted by Fathmawati et al. (2017) in Indonesia. The findings demonstrated (after adjusting for smoking history, age, and family history of cancer) an association between prolonged exposure (more than 10 years) to nitrate concentration in drinking water above 11.29 mg/L of nitrate as N, with an increased risk of colorectal cancer occurrence (OR = 4.31; 95% CI: 1.32–14.09). Moreover, in a study conducted in Iowa, long-term ingestion (\geq 4 years) of elevated nitrate in drinking water (>5 mg/L of nitrate as N) was associated with an increased risk of bladder cancer among postmenopausal women, after adjusting for covariates such as smoking status and total trihalomethane levels (HR = 1.62; 95% CI: 1.06, 2.47; Jones et al., 2016). Another study conducted in Iowa and controlled for confounders like trihalomethane levels demonstrated that high nitrate levels (>5 mg/L of nitrate as N) in public water supplies were associated with an increased risk of renal cancer (HR = 2.3, 95% CI:1.2–4.3; Jones et al., 2017).

Additionally, nitrate toxicity to laboratory animals has been reported in several studies (Alonso & Camargo, 2003; Camargo et al., 2005; Tsai & Chen, 2002). For example, long-term exposure of freshwater invertebrates to nitrate concentration of 10 mg/L resulted in adverse health outcomes in these animals Carmargo et al. (2005).

4.1. Strengths and Limitations

This ecological study by design has the advantage of a relatively large health data set and controlled for the counties socioeconomic status and the two major waterborne agrichemicals used in Nebraska. However, because of the study design, individual-level exposures to drinking water were not studied and other cancer risk factors (including other water contaminants) were not accounted for. Instead, we assumed that the agrichemical concentrations in monitoring wells are the same in drinking water, thereby captured true exposure. Additionally, the use of secondary data and the approach to investigate subtypes of pediatric cancers at the county level did not allow



Figure 7. (a) Age-adjusted incidence (AAI) of pediatric lymphoma per county in Nebraska from 1987 to 2016. (b) Age-adjusted incidence (AAI) of pediatric lymphoma in Nebraska counties compared to the national average.



Multivariable Analysis of the Associations Between Pediatric Cancers (Brain and Other CNS, Leukemia, and Lymphoma) and Agrichemical Concentrations in Groundwater (Atrazine and Nitrate)

	Brain and other CNS full model ^a	Leukemia full model ^a	Lymphoma full model ^a
Variables	IRR _a (95% CI)	IRR _a (95% CI)	IRR _a (95% CI)
Atrazine GW ^b			
Group 1	Reference	Reference	Reference
Group 2	6.46 (6.35–6.57)	3.90 (3.83–3.96)	7.81 (7.67–7.95)
Group 3	11.20 (11.01–11.39)	6.28 (6.18–6.38)	12.76 (12.53–13.00)
Group 4	13.50 (13.26–13.74)	6.22 (6.12–6.32)	12.79 (12.55–13.04)
Nitrate GW ^b			
Group 1	Reference	Reference	Reference
Group 2	8.39 (8.24–8.54)	7.35 (7.22–7.48)	5.59 (5.48–5.69)
Group 3	0.48 (0.47–0.49)	0.51 (0.50–0.52)	0.39 (0.38–0.40)
Group 4	0.83 (0.81–0.85)	0.66 (0.65–0.68)	0.62 (0.60–0.63)
Socioeconomic status (RPL_Theme1)	4.88 (4.82–4.93)	6.34 (6.27–6.41)	3.02 (2.98-3.05)

^aNegative binomial multivariable analysis performed. IRR_a = Incidence rate ratio adjusted. ^bAtrazine or nitrate concentration in groundwater.

detailed longitudinal analysis. Moreover, the authors assumed that all the children diagnosed with cancer during the study period had similar exposure regarding nitrate and atrazine concentrations. We also assumed that the county of residence at the time of diagnosis was the county where the exposure occurred.

5. Conclusions

In this study, the authors determined the mean atrazine and nitrate concentrations and the age-adjusted pediatric cancer incidences in each county in Nebraska from 1987 to 2016. The age-adjusted incidence of pediatric brain and other CNS tumors was higher than the national average in 63% of the Nebraska counties. The authors also examined the relationship between atrazine concentrations, nitrate concentrations, and pediatric cancers for the three most prevalent pediatric cancers in Nebraska (brain and other CNS, leukemia, and lymphoma). An association was found between relatively higher atrazine or nitrate concentration and an increased incidence rate of pediatric cancers (brain and other CNS, leukemia, and lymphoma). Nebraska is dominated by industrial agriculture; these results do not necessarily prove a causal relationship but suggest that the use of agrichemicals such as atrazine and nitrate poses a significant threat to pediatric health regarding brain and other CNS cancers, leukemia, and lymphoma occurrence. Further research is recommended to validate these findings, such as a case-control study to measure individual-level exposure and other potential confounders.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

The following data sets used for this research are publicly available. Links and parameters are provided below and complete references are given in the References section: (a) The 2010 US decennial census data for Nebraska county populations were obtained from the National Historical Geographic Information System database. https://www.nhgis.org/; accessed 07 October 2020. The following parameters were used to extract the data: the geographic level was set to county, the decennial census year 2010 was selected for years, the topics included age and the data sets were set to 2010_SF1a. (b) Atrazine and nitrate data were retrieved from the water quality portal (https://www.waterqualitydata.us/portal/; accessed 7 October 2020) and Quality-assessed Agrichemical Contaminant Database for Nebraska Groundwater (https://clearinghouse.nebraska.gov/well-explorer2020;accessed 7 October 2020). The parameters set for the water quality portal are as follows: For the location filters, the



Country was set to United States of America, the State to Nebraska, All counties and all sites were selected. All data sources (USGS, STEWARDS, EPA) were selected, the date ranges were set from 01/01/1987 to 12/31/2016; water (NWIS, STEWARDS, and STORET) was selected for sample media. Nutrients and Pesticides were selected for the Characteristic group. Regarding the download options we picked the site data only and Comma-Separated. For the Nebraska Groundwater Quality Clearinghouse, the filter was set to Domestic for Well type. (c) The Nebraska state and county boundary shapefiles were extracted from the United States Census Bureau https://www2.census.gov/geo/tiger/TIGER2019/; accessed 7 October 2020. The Files named County and State were selected and downloaded separately. (d) The CDC SVI is available at https://www.atsdr.cdc.gov/placeand-health/svi/data_documentation_download.html. For parameters chosen, the year was set to 2016, Geography to Nebraska; Geography type to Counties and File type to CVS file. Additional data (pediatric cancer data) supporting this research is not available to the public or research community unless access is granted by the Nebraska DHHS upon completion and approval of data use agreement forms. For further information contact the Nebraska DHHS.PublicRecords@nebraska.gov.

Acknowledgments

The authors wish to extend their gratitude to Sammy's Superheroes Foundation for funding this project and the Nebraska DHHS for granting access to the pediatric cancer data.

References

- Agency for Toxic Substances and Disease Registry (ATSDR). (2016). CDC SVI documentation 2016. Retrieved from https://www.atsdr.cdc.gov/placeandhealth/svi/data_documentation_download.html
- Agopian, A. J., Lupo, P. J., Canfield, M. A., & Langlois, P. H. (2013). Case–Control study of maternal residential atrazine exposure and male genital malformations. *American Journal of Medical Genetics, Part A*, 161(5), 977–982. https://doi.org/10.1002/ajmg.a.35815
- Alonso, A., & Camargo, J. A. (2003). Short-term toxicity of ammonia, nitrite, and nitrate to the aquatic snail Potamopyrgus antipodarum (Hydrobiidae, Mollusca). Bulletin of Environmental Contamination and Toxicology, 70(5), 1006–1012. https://doi.org/10.1007/s00128-003-0082-5 American Childhood Cancer Organization (ACCO). (2014). Special section: Cancer in children & adolescents.Retrieved from https://www.acco.
- org/wp-content/uploads/2014/11/ACS-Special-Report-2014.pdf Aschebrook-Kilfoy, B., Heltshe, S. L., Nuckols, J. R., Sabra, M. M., Shuldiner, A. R., Mitchell, B. D., et al. (2012). Modeled nitrate levels in well
- water supplies and prevalence of abnormal thyroid conditions among the Old Order Amish in Pennsylvania. *Environmental Health*, 11(1), 6. https://doi.org/10.1186/1476-069x-11-6
- Boffetta, P., Adami, H. O., Berry, C., & Mandel, J. S. (2013). Atrazine and cancer: A review of the epidemiologic evidence. *European Journal of Cancer Prevention*, 22(2), 169–180. https://doi.org/10.1097/cej.0b013e32835849ca
- Booth, B. J., Ward, M. H., Turyk, M. E., & Stayner, L. T. (2015). Agricultural crop density and risk of childhood cancer in the midwestern United States: An ecologic study. *Environmental Health*, 14(1), 82. https://doi.org/10.1186/s12940-015-0070-3
- Brender, J. D., & Weyer, P. J. (2016). Agricultural compounds in water and birth defects. *Current Environmental Health Reports*, 3(2), 144–152. https://doi.org/10.1007/s40572-016-0085-0
- Camargo, J. A., Alonso, A., & Salamanca, A. (2005). Nitrate toxicity to aquatic animals: A review with new data for freshwater invertebrates. *Chemosphere*, 58(9), 1255–1267. https://doi.org/10.1016/j.chemosphere.2004.10.044
- Carozza, S. E., Li, B., Elgethun, K., & Whitworth, R. (2008). Risk of childhood cancers associated with residence in agriculturally intense areas in the United States. *Environmental Health Perspectives*, 116(4), 559–565. https://doi.org/10.1289/ehp.9967
- Clegg, L. X., Reichman, M. E., Miller, B. A., Hankey, B. F., Singh, G. K., Lin, Y. D., et al. (2009). Impact of socioeconomic status on cancer incidence and stage at diagnosis: Selected findings from the surveillance, epidemiology, and end results: National longitudinal mortality study. *Cancer Causes & Control*, 20(4), 417–435. https://doi.org/10.1007/s10552-008-9256-0
- Corley, B., Bartelt-Hunt, S., Rogan, E., Coulter, D., Sparks, J., Baccaglini, L., et al. (2018). Using watershed boundaries to map adverse health outcomes: Examples from Nebraska. *Environmental Health Insights*, 12. https://doi.org/10.1177/1178630217751906
- DeSimone, L. A., Hamilton, P. A., & Gilliom, R. J. (2009). Quality of water from domestic wells in principal aquifers of the United States, 1991–2004—overview of major findings. US Geological Survey Circular, 1332, 48.
- Du, S. T., Zhang, Y. S., & Lin, X. Y. (2007). Accumulation of nitrate in vegetables and its possible implications to human health. Agricultural Sciences in China, 6(10), 1246–1255. https://doi.org/10.1016/s1671-2927(07)60169-2
- Environmental Systems Research Institute (ESRI). (2019). ArcGIS ProArcGIS Pro 2.4. Retrieved from https://arcgis.pro/arcgis-pro-2-4/
- Environmental Systems Research Institute (ESRI) ArcGIS Pro. (2020). Data classification methods. Retrieved from https://pro.arcgis.com/en/ pro-app/help/mapping/layer-properties/data-classification-methods.htm
- Fan, W., Yanase, T., Morinaga, H., Gondo, S., Okabe, T., Nomura, M., et al. (2007). Atrazine-induced aromatase expression is SF-1 dependent: Implications for endocrine disruption in wildlife and reproductive cancers in humans. *Environmental Health Perspectives*, 115(5), 720–727. https://doi.org/10.1289/ehp.9758
- Fathmawati, Fachiroh, J., Gravitiani, E., Sarto, & Husodo, A. H. (2017). Nitrate in drinking water and risk of colorectal cancer in Yogyakarta, Indonesia. Journal of Toxicology and Environmental Health, Part A, 80(2), 120–128. https://doi.org/10.1080/15287394.2016.1260508
- Freeman, L. E. B., Rusiecki, J. A., Hoppin, J. A., Lubin, J. H., Koutros, S., Andreotti, G., et al. (2011). Atrazine and cancer incidence among pesticide applicators in the agricultural health study (1994–2007). Environmental Health Perspectives, 119(9), 1253–1259. https://doi.org/10.1289/ ehp.1103561
- Garcia-Gil, M., Elorza, J. M., Banque, M., Comas-Cufi, M., Blanch, J., Ramos, R., et al. (2014). Linking of primary care records to census data to study the association between socioeconomic status and cancer incidence in Southern Europe: A nation-wide ecological study. *PLoS One*, 9(10), e109706. https://doi.org/10.1371/journal.pone.0109706
- Gely-Pernot, A., Saci, S., Kernanec, P. Y., Hao, C., Giton, F., Kervarrec, et al. (2017). Embryonic exposure to the widely-used herbicide atrazine disrupts meiosis and normal follicle formation in female mice. *Scientific Reports*, 7(1), 1–14. https://doi.org/10.1038/s41598-017-03738-1
- Holtby, C. E., Guernsey, J. R., Allen, A. C., VanLeeuwen, J. A., Allen, V. M., & Gordon, R. J. (2014). A population-based case-control study of drinking-water nitrate and congenital anomalies using geographic information systems (GIS) to develop individual-level exposure estimates. International Journal of Environmental Research and Public Health, 11(2), 1803–1823. https://doi.org/10.3390/ijerph110201803

- Horzmann, K. A., Reidenbach, L. S., Thanki, D. H., Winchester, A. E., Qualizza, B. A., Ryan, G. A., et al. (2018). Embryonic atrazine exposure elicits proteomic, behavioral, and brain abnormalities with developmental time specific gene expression signatures. *Journal of Proteomics*, 186, 71–82. https://doi.org/10.1016/j.jprot.2018.07.006
- IARC Working Group on the Evaluation of Carcinogenic Risks to Humans. (2010). Ingested nitrate and nitrite, and cyanobacterial peptide toxins IARC. *Monogr Eval Carcinog Risks Hum*, 94, 1–412.
- IBM Corp. (2019). IBM SPSS statistics for windows, version 26.0. IBM Corp. Retrieved from https://www.ibm.com/products/spss-statistics
 - Jones, R. R., Weyer, P. J., DellaValle, C. T., Inoue-Choi, M., Anderson, K. E., Cantor, K. P., et al. (2016). Nitrate from drinking water and diet and bladder cancer among postmenopausal women in Iowa. *Environmental Health Perspectives*, 124(11), 1751–1758. https://doi.org/10.1289/ ehp191
 - Jones, R. R., Weyer, P. J., DellaValle, C. T., Robien, K., Cantor, K. P., Krasner, S., et al. (2017). Ingested nitrate, disinfection by-products, and kidney cancer risk in older women. *Epidemiology*, 28(5), 703–711. https://doi.org/10.1097/ede.00000000000647
 - Malagoli, C., Costanzini, S., Heck, J. E., Malavolti, M., De Girolamo, G., Oleari, P., et al. (2016). Passive exposure to agricultural pesticides and risk of childhood leukemia in an Italian community. *International Journal of Hygiene and Environmental Health*, 219(8), 742–748. https:// doi.org/10.1016/j.ijheh.2016.09.015
 - Manson, S., Schroeder, J., Van Riper, D., Kugler, T., & Ruggles, S. (2021). IPUMS national historical geographic information system: Version 16.0 [dataset]. IPUMS. https://doi.org/10.18128/D050.V16
 - Mensinga, T. T., Speijers, G. J., & Meulenbelt, J. (2003). Health implications of exposure to environmental nitrogenous compounds. *Toxicological Reviews*, 22(1), 4141–5151. https://doi.org/10.2165/00139709-200322010-00005
 - Monti, M. M., David, F., Shin, M., & Vaidyanathan, A. (2019). Community drinking water data on the national environmental public health tracking network: A surveillance summary of data from 2000 to 2010. Environmental Monitoring and Assessment, 191(9), 557. https://doi. org/10.1007/s10661-019-7710-y
 - National Cancer Institute. (2017). State cancer profiles. Retrieved from https://statecancerprofiles.cancer.gov/quick-profiles/index.php?state name=nebraska
 - National Cancer Institute. (2021). Surveillance, epidemiology and end results program calculating age-adjusted rates. Retrieved from https://seer. cancer.gov/seerstat/tutorials/aarates/step1.html
 - National Water Quality Monitoring Council. (2021). Water quality portal. Retrieved from https://www.waterqualitydata.us/portal/
 - Nebraska Department of Agriculture and USDA NASS. (2020). Nebraska field office Nebraska agriculture fact card. Retrieved from https://nda.nebraska.gov/facts.pdf
 - Nebraska Department of Agriculture. (ND). Nebraska agriculture. Retrieved from https://nda.nebraska.gov/publications/ne_ag_facts_brochure.pdf
 - Nebraska Department of Environmental Quality (NDEQ). (2020). 2020 Nebraska groundwater quality monitoring report. Retrieved from http:// dee.ne.gov/Publica.nsf/PubsForm.xsp?documentId=C3C47F71DBDA83338625863100728C51&action=openDocument
 - NIH National Cancer Institute. (2020). Childhood cancers. Retrieved from https://www.cancer.gov/types/childhood-cancers#types
 - Nolan, B. T., & Stoner, J. D. (2000). Nutrients in groundwaters of the conterminous United States, 1992–1995. Environmental Science & Technology, 34(7), 1156–1165. https://doi.org/10.1021/es9907663
 - Rhoades, M. G., Meza, J. L., Beseler, C. L., Shea, P. J., Kahle, A., Vose, J. M., et al. (2013). Atrazine and nitrate in public drinking water supplies and non-Hodgkin lymphoma in Nebraska (Vol. 7). Environmental health insights. EHI-S10629. https://doi.org/10.4137/ehi.s10629 Robin, N. H., & Farmer, M. (2017). Pediatric cancer genetics. Elsevier Health Sciences.
 - Rusiecki, J. A., De Roos, A., Lee, W. J., Dosemeci, M., Lubin, J. H., Hoppin, J. A., et al. (2004). Cancer incidence among pesticide applicators exposed to atrazine in the agricultural health study. *Journal of the National Cancer Institute*, 96(18), 1375–1382. https://doi.org/10.1093/jnci/ dih264
 - Sathiakumar, N., MacLennan, P. A., Mandel, J., & Delzell, E. (2011). A review of epidemiologic studies of Triazine herbicides and cancer. Critical Reviews in Toxicology, 41(sup1), 1–34. https://doi.org/10.3109/10408444.2011.554793
 - Thorpe, N., & Shirmohammadi, A. (2005). Herbicides and nitrates in groundwater of Maryland and childhood cancers: A geographic information systems approach. *Journal of Environmental Science and Health Part C*, 23(2), 261–278. https://doi.org/10.1080/10590500500235001
 - Tsai, S. J., & Chen, J. C. (2002). Acute toxicity of nitrate on Penaeus monodon juveniles at different salinity levels. *Aquaculture*, 213(1-4), 163–170. https://doi.org/10.1016/s0044-8486(02)00023-6
 - United States Census Bureau (2019). Tiger. Retrieved from https://www2.census.gov/geo/tiger/TIGER2019/
 - United States Environmental Protection Agency (EPA) (2007). Nitrates and nitrites-EPA archives. Retrieved from https://archive.epa.gov/ region5/teach/web/pdf/nitrates_summary.pdf
 - United States Environmental Protection Agency (EPA). (2020). Ground water and drinking water. *National primary drinking water regulations*. Retrieved from https://www.epa.gov/ground-water-and-drinking-water/national-primary-drinking-water-regulations#Inorganic
 - Unites States Environmental Protection Agency (EPA). (2020). Private drinking water wells. Retrieved from https://www.epa.gov/privatewells
 - University of Nebraska-Lincoln. (2020). Quality-assessed agrichemical contaminant database for Nebraska ground water. A cooperative project of the Nebraska departments of agriculture, environmental quality, and natural resources and the University of Nebraska Lincoln. Retrieved from https://clearinghouse.nebraska.gov/well-explorer
 - U.S. Cancer Statistics Working Group. (2020). U.S. Cancer statistics data visualizations tool, based on 2019 submission data (1999-2017). U.S. Department of Health and Human Services, Centers for Disease Control and Prevention and National Cancer Institute. Retrieved from www.cdc.gov/cancer/dataviz
 - USDA ERS United States Department of Agriculture Economic Research Service. (2019). Fertilizer use and price. Retrieved from https://www.ers.usda.gov/data-products/fertilizer-use-and-price/
 - Waller, S. A., Paul, K., Peterson, S. E., & Hitti, J. E. (2010). Agricultural-related chemical exposures, season of conception, and risk of gastroschisis in Washington State. American Journal of Obstetrics and Gynecology, 202(3), 241–e1. https://doi.org/10.1016/j.ajog.2010.01.023
- Ward, M. H., Jones, R. R., Brender, J. D., De Kok, T. M., Weyer, P. J., Nolan, B. T., et al. (2018). Drinking water nitrate and human health: An updated review. International Journal of Environmental Research and Public Health, 15(7), 1557. https://doi.org/10.3390/ijerph15071557
- Weber, G. J., Sepúlveda, M. S., Peterson, S. M., Lewis, S. S., & Freeman, J. L. (2013). Transcriptome alterations following developmental atrazine exposure in zebrafish are associated with disruption of neuroendocrine and reproductive system function, cell cycle, and carcinogenesis. *Toxicological Sciences*, 132(2), 458–466. https://doi.org/10.1093/toxsci/kft017
- Wieben, C. M. (2019). Estimated annual agricultural pesticide use by major crop or crop group for states of the conterminous United States, 1992–2017, version 2.0, May 2020; data release. US Geological Survey. https://doi.org/10.5066/P9HHG3CT